Author's response to reviews

Title: Common variation in EMSY and risk of breast and ovarian cancer: a case-control study using HapMap tagging SNPs

Authors:

Patrick R Benusiglio (pbenusiglio@yahoo.com)
Fabienne Lesueur (lesueur@cng.fr)
Craig Luccarini (craig.luccarini@srl.cam.ac.uk)
Joan McIntosh (joan.mcintosh@srl.cam.ac.uk)
Robert N Luben (robert.luben@srl.cam.ac.uk)
Paula Smith (paula.smith@srl.cam.ac.uk)
Alison Dunning (alison.dunning@srl.cam.ac.uk)
Douglas F Easton (douglas.easton@srl.cam.ac.uk)
Bruce AJ Ponder (baip@mole.bio.cam.ac.uk)
Paul D Pharoah (paul.pharoah@srl.cam.ac.uk)

Version: 4 Date: 18 July 2005

Author's response to reviews:

The requested formatting changes were as follows:

Authors' list- Please provide the e-mail addresses for the authors separately form the author list.
Done.

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country.
Done.

Materials & Methods - Please rename this section 'Methods'.
Done.

Tables - These should be oriented in portrait rather than in landscape.
Done.

Footnotes - We are unable to accommodate footnotes in the full text version of the manuscript.
Footnotes have been removed. When necessary, the information has been added to the table legend.

Figures- The figures should be cropped to minimise white space around the image.
White space around the figure has been minimised.

Best regards,

P Benusiglio